Author’s response to reviews

Title: LONGITUDINAL STUDY OF GASTROESOPHAGEAL REFLUX AND EROSIVE TOOTH WEAR

Authors:

Clive Wilder-Smith (cws@braingut.com)
Andrea Materna (lab@foointolerances.org)
Lukas Martig (lukasmartig@gmail.com)
Adrian Lussi (adrian.lussi@unibe.ch)

Version: 2 Date: 11 Jul 2017

Author’s response to reviews:

Dear Editors and reviewers

Thank you for the comments and suggestions, which we have incorporated in the attached version of the paper. This is clearly an observational study. The study indexed in clinicaltrials.gov is the follow-on study for up to 500 patients, but is also listed as observational.

Thank you for your constructive and helpful input. We hope with these changes the paper is acceptable for publication.

Kind regards

Clive Wilder-Smith